Author's response to reviews

Title: Treatment of the Diabetic Foot - To amputate or not?

Authors:

Elroy P Weledji (elroypat@yahoo.co.uk)
Pius k Fokam (piuskamus@yahoo.com)

Version: 7  Date: 8 October 2014

Author's response to reviews:

COVER LETTER: Treatment of diabetic foot: To amputate or not?

Dear Editor,

I send to you a revised manuscript of the paper with corrections that meet with the guidelines of the journal.

(1) I have corrected the authors contribution as requested explaining the role of the participants

(2) A statement confirming that informed consent was obtained from the patient for the diabetic foot image to be published is inserted under ‘Acknowledgment’

Thank you

Dr EP Weledji Corresponding author
8/10/14